Author's response to reviews

Title: Occupation recorded on certificates of death compared with self-report: the Atherosclerosis Risk in Communities (ARIC) Study

Authors:

Aurelian Bidulescu (abidulescu@msm.edu)
Kathryn M Rose (kathryn_rose@unc.edu)
Susanne H Wolf (susanne_wolf@unc.edu)
Wayne D Rosamond (wayne_rosamond@unc.edu)

Version: 3 Date: 17 August 2007

Author's response to reviews:

August 17, 2007

The BioMed Central Editorial Production Team
BMC Series Journals

Manuscript Number: 6919618041429035

Title: Occupation recorded on certificates of death compared with self-report: the Atherosclerosis Risk in Communities (ARIC) Study

Dear Editorial Production Team Members:

We are glad to know that our paper entitled "Occupation recorded on certificates of death compared with self-report: the Atherosclerosis Risk in Communities (ARIC) Study" has been accepted for publication in BMC Public Health. We have incorporated into the final version the formatting changes requested. Specifically, we made sure that all the capital letters used are appropriate (we maintained the upper-case letters for the study name). We have removed the text 'Authors:' from the title page and used regular font style instead of italic for the author list. We have also included the country in the affiliation details. We eliminated the full-stops from the table titles.

We hope that with these changes now incorporated our manuscript is ready for publication. We are proud to publish in BMC Public Health, a journal actively constructing an international space of high scientific level for scholarly communications.

Sincerely,

Aurelian Bidulescu, MD, PhD, MPH
Assistant Professor